Author’s response to reviews

Title: Knee arthrodesis versus above-the-knee amputation after septic failure of revision total knee arthroplasty: comparison of functional outcome and complication rates.

Authors:

Sven Hungerer (sven.hungerer@bgu-murnau.de)

Martin Kiechle (martin.kiechle@bgu-murnau.de)

Christian von Rüden (Christian.vonRueden@bgu-murnau.de)

Matthias Militz (matthias.militz@bgu-murnau.de)

Knut Beitzel (beitzelknut@hotmail.com)

Mario Morgenstern (morgenstern.mario@googlemail.com)

Version: 2 Date: 27 Oct 2017

Author’s response to reviews:

ad 1. The ethical approval was obtained from the ethical board of the Bavarian Medical Association. Patients gave their written consent for the analysis of the data for scientific research.

This was added and highlighted in the manuscript. If required we could add the letter of the ethical board as supplementary material.

ad 2. in the contribution section MM (Matthias Militz) was changed to MM1 and Mario Morgenstern to MM2 as suggested.

Best regards